Author's response to reviews

Title: Mobility of primary health workers in China

Authors:

Qingyue Meng Prof (qmeng@sdu.edu.cn)
Jing Yuan Dr (yuanjing@mail.sdu.edu.cn)
Limei Jing Ms (limei100409@mail.sdu.edu.cn)
Junhua Zhang Dr (jzhang70@yahoo.com)

Version: 5 Date: 12 March 2009

Author's response to reviews:

1. All references have been numbered within brackets.
2. In the references section, references are numbered accordingly.
3. In the title page, English name of the organization for Dr Zhang Junhua (co-author) was modified as suggested by Dr Zhang Junhua.